Author's response to reviews

Title: Acute effect of meal glycemic index and glycemic load on blood glucose and insulin responses in humans

Authors:

Jose J Galgani (jgalgani@inta.cl)
Carolina C Aguirre (caguirre@inta.cl)
Erik E Diaz (ediaz@inta.cl)

Version: 7 Date: 4 September 2006

Author's response to reviews:

Line numbers - Please remove.
DONE

Competing interests - If there are none to declare, please write only The authors declare that they have no competing interests.
DONE

Qualifications - Please remove the authors' qualifications (e.g. PhD / MD) from the title page.
DONE

Country - Please remember to include the country in the affiliation details.
DONE

References - References should be cited in the text using consecutive numbers in square brackets e.g. [1].
DONE

Figure legends/tables - Please insert these after the references.
DONE

Figures - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure cropping guide may be of some help, and can be viewed here: http://www.biomedcentral.com/info/ifora/cropping.
DONE